Author's response to reviews

Title: Translation elongation factor eEF1A2 is a potential oncoprotein that is overexpressed in two-thirds of breast tumors

Authors:

Victoria A.L. Tomlinson (V.A.L.Tomlinson@sms.ed.ac.uk)
Helen J. Newbery (h.newbery@ed.ac.uk)
Naomi R. Wray (Naomi.Wray@ed.ac.uk)
Juliette Jackson (Juliettemurray@doctors.org.uk)
Alexey Larionov (alexey@larionov.co.uk)
William R. Miller (W.R.Millar@ed.ac.uk)
J. MICHAEL Dixon (mdixon@staffmail.ed.ac.uk)
Catherine M. Abbott (C.Abbott@ed.ac.uk)

Version: 5 Date: 31 August 2005

Author's response to reviews:

All minor editorial changes requested have been made, and figures cropped according to guidelines.